# Forced Changes in the Arctic Freshwater Budget Emerge in the Early 21<sup>st</sup> Century

# Alexandra Jahn<sup>1</sup>, Rory Laiho<sup>1</sup>

<sup>1</sup>Department of Atmospheric and Oceanic Sciences and Institute of Arctic and Alpine Research,
 University of Colorado Boulder, 450 UCB, Boulder, 80309-0450, USA.

## Key Points:

1

2

3

6

13

7	• The observed increase in Arctic liquid freshwater (FW) storage is likely already
8	driven by climate change
9	• A forced change in liquid FW flux through Nares Strait is likely to emerge within
10	the next decade
11	• The already changing nature of many FW budget terms can delay detection of shift
12	and emergence from observations

Manuscript under review in GRL after revisions. July 15 2020.

Corresponding author: Alexandra Jahn, alexandra.jahn@colorado.edu

#### 14 Abstract

Arctic liquid freshwater (FW) storage has shown a large increase over the past decades, 15 posing the question: Is the Arctic FW budget already showing clear signs of anthropogenic 16 climate change, or are the observed changes the result of multi-decadal variability? We 17 show that the observed change in liquid and solid Arctic FW storage is likely already 18 driven by the changing climate, based on ensemble simulations from a state-of-the-art 19 climate model. Generally, the emergence of forced changes in Arctic FW fluxes occurs 20 earlier for oceanic fluxes than for atmospheric or land fluxes. Nares Strait liquid FW flux 21 is the first to show emergence outside the range of background variability, with this change 22 potentially already occurring. Other FW fluxes have likely started to shift but have not 23 yet emerged into a completely different regime. Future emissions reductions have the po-24 tential to avoid the emergence of some FW fluxes beyond the background variability. 25

26

#### Plain Language Summary

The surface waters of the Arctic Ocean are fresher than the rest of the world oceans, 27 due to the input of large amounts of river runoff. The very fresh surface ocean affects 28 the ocean circulation and climate not just in the Arctic Ocean, but also at lower lati-29 tudes, especially in the North Atlantic. The last two decades have seen a freshening of 30 the surface Arctic Ocean, for reasons that are currently unknown. Here we demonstrate 31 that this freshening is likely already driven by climate change. Furthermore, we find that 32 due to man-made climate change, Arctic freshwater fluxes to the North Atlantic are also 33 likely to soon start showing signs of change beyond the range of the variability we have 34 observed in the past. The information provided here about the expected timing of the 35 emergence of climate change signals will allow us to monitor upcoming changes in real 36 time, to better understand how changes in the Arctic Ocean can impact climate world-37 wide. 38

#### <sup>39</sup> 1 Introduction

Arctic Ocean liquid freshwater (FW) storage has shown a large increase from the 1990s until at least 2014 (e.g., Proshutinsky et al., 2009; Rabe et al., 2011, 2014; McPhee et al., 2009; Giles et al., 2012; Wang et al., 2019). Recent work suggests that this Arcticwide increase is likely driven primarily by natural modes of variability rather than by anthropogenic climate change (Johnson et al., 2018). This contrasts with the observed

reduction in the solid Arctic FW storage in sea ice over the same period, which has been 45 shown to be at least partially driven by climate change (e.g., Notz & Marotzke, 2012; 46 Wang et al., 2019; Schweiger et al., 2019). Furthermore, climate models predict a 21<sup>st</sup> 47 century increase in the liquid FW storage and in many Arctic FW fluxes (e.g., Holland 48 et al., 2006, 2007; Koenigk et al., 2007; Vavrus et al., 2012; Haine et al., 2015; Shu et al., 49 2018). While some Arctic FW fluxes have started to show changes in line with these pre-50 dictions, others have not. In particular, as expected for a warmer climate, the Bering 51 Strait FW influx (Woodgate, 2018), river runoff (Peterson et al., 2006), and net precip-52 itation (Haine et al., 2015) have all increased, and solid FW storage has decreased (e.g., 53 Haine et al., 2015; Wang et al., 2018). However, the liquid FW exports from the Arc-54 tic have not yet shown any clear changes or trends (de Steur et al., 2009; Curry et al., 55 2014; de Steur et al., 2018; Haine et al., 2015). This poses the question as to when the 56 current monitoring of Arctic FW storage and fluxes will be able to detect an anthropogenic 57 climate change signal. 58

Attributing observed change in the Arctic FW budget terms either to natural modes 59 of variability or climate change is challenging due to the combination of the known in-60 fluence of decadal to multi-decadal modes of variability on Arctic FW (e.g., Proshutin-61 sky & Johnson, 1997; Proshutinsky et al., 2002; Polyakov et al., 2008; Johnson et al., 2018) 62 and the short (25 years or less) continuous records available for many of the Arctic FW 63 budget terms, in particular the liquid oceanic FW fluxes (e.g., de Steur et al., 2018; Münchow, 64 2016; Curry et al., 2014; Rabe et al., 2009, 2014). Here we show when we can expect to 65 detect anthropologically-forced changes in the various Arctic freshwater budget terms, 66 by determining the time of emergence outside the background variability using climate 67 model simulations. Furthermore, we assess whether the detection of a forced change is 68 dependent upon future emissions pathway choices. To separate the forced change from 69 natural variability on multiple timescales and between emissions scenarios, we use en-70 semble simulations over the 20<sup>th</sup> and 21<sup>st</sup> centuries from a fully-coupled state-of-the-art 71 earth system model under two different forcing scenarios (Kay et al., 2015; Sanderson 72 et al., 2017). We find that the time of emergence of forced changes varies widely amongst 73 Arctic FW budget terms. Some are already showing a climate-change signal or are likely 74 to do so soon, in particular the FW storage terms and the Nares and Davis Strait liq-75 uid FW fluxes. 76

-3-

#### 77 2 Methods

78

#### 2.1 Model and Simulations

To assess the time of emergence of a climate change signal beyond natural inter-79 nal variability in the Arctic FW budget, we use the Community Earth System Model 80 (CESM) Large Ensemble (CESM LE; Kay et al., 2015) and a companion ensemble, the 81 CESM Low Warming ensemble (CESM LW; Sanderson et al., 2017). Both ensembles use 82 the CESM1.1, a fully-coupled, state-of-the-art global earth system model (Hurrell et al., 83 2013), and differ only in the applied forcing for the 21st century, allowing us to assess 84 whether any of the results are sensitive to different future emissions choices. The his-85 torical CESM LE ensemble is created in 1920 through round-off level perturbations to 86 the temperature field (Kay et al., 2015). After 2006, the CESM LE uses the high-emissions 87 RCP8.5 scenario, leading to over 4 °C warming by 2100. The CESM LW is branched from 88 the CESM LE ensemble members in 2006 and uses the RCP8.5 forcing until 2017, at which 89 point it is then forced by a reduced emission scenario designed so that global warming 90 stabilizes at 2 °C for several decades before the end of the 21<sup>st</sup> century (Sanderson et al., 91 2017). The background variability is determined from the 1800 year long pre-industrial 92 control simulation from the CESM LE project. 93

To consistently compare the CESM LE and LW, despite their different ensemble sizes (40 versus 11, respectively), all results shown are from the first 11 ensemble members of the CESM LE, referred to in the following as CESM LE. The effect of using the 40-member CESM LE was assessed and is discussed where applicable (with relevant figures in the supplementary), to provide insights into the impact of larger internal variability. None of the main conclusions are impacted by the use of the 11-member versus the 40-member CESM LE.

The CESM1.1 has already been used for a wide range of Arctic climate studies and 101 generally performs well in the Arctic (e.g., Barnhart et al., 2015; Swart et al., 2015; DeRe-102 pentigny et al., 2016; Jahn et al., 2016; Jahn, 2018; Auclair & Tremblay, 2018; Morri-103 son et al., 2019; England et al., 2019; Smith & Jahn, 2019). Nonetheless, as all models, 104 the CESM1.1 has some biases. In terms of the simulated Arctic FW budget, which is 105 calculated relative to the commonly used reference salinity of 34.8 (Aagaard & Carmack, 106 1989; Serreze et al., 2006; Haine et al., 2015), those biases are found primarily in the FW 107 exports from the Arctic Ocean (see Fig. 1a for the ocean gateways used here). The liq-108

-4-

uid FW exports are underestimated by the model while the solid FW exports are over-109 all too large, due to too much FW residing in the sea ice relative to the ocean over the 110 observational period (see the Supplementary Material section S1 for details on the cal-111 culation of the FW budget and Table S1 for a comparison with observations). In par-112 ticular, the Fram Strait and Barrow Strait liquid FW export are underestimated almost 113 by a factor of three by the model for the late 20<sup>th</sup> century, while the BSO liquid FW ex-114 port is nearly 10 times as large as observed. However, the overall exchange of FW with 115 the North Atlantic is within the observational uncertainty range. Furthermore, none of 116 the biases found in the CESM1.1 Arctic FW budget are unique to the CESM1.1. Both 117 fully-coupled CMIP3 (Holland et al., 2007) and CMIP5 models (Shu et al., 2018) as well 118 as reanalysis-forced regional and global ocean-sea ice models (Jahn et al., 2012; Wang 119 et al., 2016b, 2016a) exhibit biases in their simulated FW exports from the Arctic. So, 120 while the results presented here have the caveat that they are derived from a single model, 121 this study presents a first assessment of the changes in the Arctic FW budget in the con-122 text of internal variability, which is only possible when using an ensemble of simulations 123 from one model. By enabling the separation of internal variability from the forced re-124 sponse, this study fills a clear gap in our understanding of the changing Arctic FW bud-125 get (as identified in Lique et al., 2016; Cornish et al., 2020). 126

127

#### 2.2 Definition of Shift and Emergence

In order to detect a climate change driven signal in the Arctic FW budget, we determine when annual-mean Arctic FW budget terms first depart from the pre-industrial natural internal variability range (*"shift years"*) and when they enter a completely different regime, with no overlap with the pre-industrial state (*"emergence years"*). While individual shift years can occur due to an unlikely extreme event (i.e., natural variability) or due to a forced change (i.e., climate change), emergence occurs only due to forced change.

To detect shift and emergence for each budget term, we use an Internal Variability Threshold (IVT) of  $\pm 3.5$  standard deviations around the mean of the 1800 year long pre-industrial control simulation. For normally-distributed processes, as most of the investigated FW fluxes are in the pre-industrial simulation,  $\pm 3.5$  standard deviations captures 99.95% of the unforced internal variability. This threshold of  $\pm 3.5$  standard deviations lies between what is known as "evidence" ( $\pm 3$  standard deviations) and "discovery" (±5 standard deviations). Smaller/larger thresholds than 3.5 standard deviations as well as non-Gaussian methods to define the IVT lead to qualitatively similar results, with some changes in the specific shifts and emergence years (see the Supplementary Material, section S2 and Fig. S3).

The *shift year* is defined as the first year in which a simulated FW term crosses the 145 pre-industrial IVT, independent of whether it subsequently crosses the IVT back into 146 the background variability. The *emergence year* is the first year when the FW term con-147 sistently stays outside the pre-industrial IVT range until the end of the simulations in 148 2100. This means that for any shift that starts before 2005, the CESM LE and CESM 149 LW shift years are the same, as they are based on the same 11 historical simulations. Emer-150 gence, however, can differ between the CESM LE and LW even before 2005, as emer-151 gence depends on the future behavior of the fluxes until 2100. Shift and emergence pe-152 riods are defined as the period between the time when the first and last ensemble mem-153 ber satisfy these criteria. Hence, the shorter the shift and emergence period, the more 154 strongly forced the simulated change is. 155

#### 156 **3 Results**

# 157

## 3.1 $21^{st}$ Century Changes in the Arctic FW Budget

Arctic FW budget terms show a large spread in how much they are projected to 158 change over the 21<sup>st</sup> century (Fig. 1b), as well as show clear differences between the low 159 and high warming scenarios by the end of the  $21^{st}$  century (referred to as "scenario im-160 pact" in the following). The largest scenario impact is seen for those FW budget terms 161 that change the most in magnitude over the 21st century, namely Arctic liquid and solid 162 FW storage, Fram Strait liquid and solid FW fluxes, the Nares Strait and Davis Strait 163 liquid FW flux, and river runoff (Fig. 1b). These changes simulated by the CESM are 164 generally consistent with those previously reported over the 21<sup>st</sup> century for different in-165 dividual models (Holland et al., 2006; Vavrus et al., 2012; Koenigk et al., 2007) as well 166 as for CMIP3 (Holland et al., 2007) and CMIP5 (Shu et al., 2018) models. Note that we 167 will focus on the larger FW budget terms, which means that except in Fig. 1b), we do 168 not show or discuss the small FW fluxes (net observed fluxes smaller than  $300 \text{ km}^3/\text{yr}$ ). 169 For completeness, plots for these fluxes are included in the Supplementary Material (Fig. S1– 170 S3). 171

-6-



**Figure 1.** Arctic domain and Climatological FW budget. (a) Ocean gateways (labeled in red) and the Arctic Ocean domain used here (shaded; showing the simulated liquid FW column [in m] over 1980–2000). BSO stands for Barents Sea Opening. (b) Climatological ensemble-mean Arctic FW budget terms for the late 20<sup>th</sup> century (1980–2000) and the late 21<sup>st</sup> century (2080-2100), with the late 21<sup>st</sup> century shown under both low warming (CESM LW; green) and high warming (CESM LE; purple). The values of the flux terms are shown on the left y-axis, the values of the FW storage terms on the right y-axis. Note that Davis Strait is shown here for reference, as it has been used in several other studies of the Arctic FW budget (e.g., Haine et al., 2015; Wang et al., 2016b, 2016a; Shu et al., 2018), but is not part of the Arctic FW budget/domain used here.

In addition to changes in the mean, we also see an increase in the variability of many oceanic liquid FW fluxes over the late 20<sup>th</sup> and 21<sup>st</sup> centuries, while the variability of the solid FW fluxes decreases as the Arctic sea ice volume/solid FW storage decreases (Fig. 2 and Fig. S2). The FW budget terms that show the largest scenario impact in the mean also show the largest scenario impact on their range of internal variability (Fig. 2 and Fig. S2).

178

#### 3.2 Shift and Emergence in Arctic FW Budget Terms

The solid and liquid Arctic FW storage terms show the earliest complete shift and 179 emergence transition of all FW terms assessed, with emergence complete in all members 180 under both scenarios by the early 2020s (Fig. 3). The very rapid emergence across all 181 ensemble members, lasting less than two decades, indicates a strongly forced change into 182 a new regime. For the liquid FW storage, the shift period is as short as the emergence 183 period for the CESM LE, again indicating a strongly forced change despite large inter-184 nal variability. However, the occurrence of a rare internal variability event outside the 185 IVT has the potential to extend the shift period. This is the case for the early  $20^{\text{th}}$  cen-186 tury start of the solid FW storage shift period: One ensemble member shows an increase 187



Figure 2. FW budget terms over time. Annual mean FW fluxes (a-h) and storage (i, j) over time for the different scenarios (CESM LE in purple, CESM LW in green, historical CESM in grey). The fluxes are labeled in the panels, with their respective  $\pm 3.5$  standard deviations IVT lines (black). The earliest shift and emergence years for each scenario are shown as vertical lines, with the shift shown as dashed light purple/green for the CESM LE/LW and emergence shown as solid dark purple/green lines for the CESM LE/LW. If the shift occurs during the historical simulation, the shift is shown as grey dashed line. Only the fluxes with observed net fluxes above 300 km<sup>3</sup> are shown here, the smaller fluxes are shown in Fig. S2.



Figure 3. Shift and Emergence Periods. Shift (thinner bars and lighter colors) and emergence (thick bars and darker colors) periods for the simulated annual mean Arctic FW variables under the historical forcing (grey), the low warming scenario CESM LW (green colors), and the high warming scenario CESM LE (purple colors). The color gradient within each bar denotes the number of members that have shifted/emerged, as indicated in the colorbar, with a focus on the edges and middle of the distribution. Note that the grey bars are by definition the same for both scenarios, as they are from the same 11 historical simulations. The impact of sampling a larger range of internal variability in the 40-member CESM LE is illustrated in Fig. S3c. It shows that while the general sequence of shift and emergence stays the same, generally longer shift periods and some longer emergence periods are found. Fig. S3c also shows the shift and emergence for the small terms of the FW budget not shown here.

- in solid FW storage above the IVT in 1943, but the forced change is towards lower solid
  FW storage and the lower bound of the IVT is not crossed until 1998 by the first ensemble member (Fig. 2j). As this early start of the shift period is due to internal variability, it disappears when a slightly larger IVT is used (Fig. S3). A similar early shift event
  occurs for the liquid FW storage when the full 40-member CESM LE is considered (Fig. S3),
  and is also due to a single crossing of the IVT in the opposite direction than the forced
  signal emerging in the 21<sup>st</sup> century.
- The Nares Strait liquid FW export from the Arctic is the next FW budget term, and the first FW flux, that shifts and then emerges from the pre-industrial background variability in all ensemble members (Fig. 3). In particular, we find that emergence already starts in the early 2000s in some ensemble members, and is complete in the early 2020s when accounting for all ensemble members of both scenarios. As for the FW storage terms, the low warming scenario does not prevent the emergence of the forced sig-

nal in the Nares Strait liquid FW export, as the forcings only begin to diverge in 2017.
However, by the end of the 21<sup>st</sup> century, the magnitude of the Nares Strait liquid FW
export increase is very different based on the two scenarios, with a much larger increase
under the high warming scenario compared to the low warming scenario (Fig. 1b and
Downstream of Nares Strait, Davis Strait exhibits similar variability as well as similar shift and emergence periods compared to Nares Strait in CESM LE and LW, but
slightly later than what is seen for Nares Strait (Fig. 2 and 3).

The Fram Strait liquid FW export also shows shift and a potential to begin to emerge 208 over the next decade (Fig. 3). Shift begins in the late 1980s, but only due to one event, 209 with the second crossing of the IVT towards larger liquid FW export not occurring un-210 til the early 2000s (Fig. 2a and Fig. 3). This indicates that the early shift event is due 211 to an anomalous, unforced event rather than climate change, similar to the observed large 212 FW export events leading to Great Salinity Anomalies in the North Atlantic (e.g., Dick-213 son et al., 1988; Belkin et al., 1998). Emergence for the Fram Strait liquid FW export 214 begins as early as the mid-2020s. However, due to the large and increasing variability 215 of the liquid Fram Strait FW export compared to the forced change (Fig. 2), the emer-216 gence period extends to 2090 (CESM LW) and beyond 2100 (CESM LE). This means 217 that full emergence outside the pre-industrial background variability may occur anytime 218 between the mid 2020s and the late  $21^{st}$  century (Fig. 3). The shorter emergence period 219 in the low warming scenario compared to that in the high warming scenario is the re-220 sult of the enhanced internal variability of the liquid Fram Strait FW export in a warmer 221 climate (Fig. 2), so that the internal variability is larger for the CESM LE than the CESM 222 LW. 223

The solid Fram Strait FW export stands out as the Arctic FW flux with the largest variability over the historical and pre-industrial period (Fig. 2 and Fig. S1). As a result, the shift period only begins in the early 2020s (Fig. 3), despite a much earlier clearly visible decrease of this FW flux within the IVT range (Fig. 2).

Both the solid Fram Strait FW export and the river runoff show a clear scenario impact on the emergence of a forced climate change signal, with emergence only occurring in the high emission scenario, but shift occurring for both the high and low warming scenarios (Fig. 3). In particular, runoff shows full emergence in the CESM LE by 2089 (and in all but one of the 40 ensemble members of the full CESM LE before 2100, see

-10-

Fig. S3c). Similarly, the solid Fram Strait FW export shows emergence in all but one ensemble member of the CESM LE by 2100 (Fig. 3). Reaching shift but not emergence under the low warming scenario means that these two FW fluxes show clear changes over time, but the fluxes do not consistently lie outside the pre-industrial IVT range by 2100 (Fig. 2).

Bering Strait and Barrow Strait liquid FW fluxes both show very long shift peri-238 ods under both scenarios, but no emergence (Fig. 3). Bering Strait liquid FW fluxes be-239 gin to shift earlier (in the 1970s) than Barrow Strait liquid fluxes (in the 2000s). How-240 ever, while over 90% of the ensemble members shift before 2100 under the high warm-241 ing scenario for both fluxes, less than 50% of the low warming scenario members shift 242 before 2100 (Fig. 3), reflecting a scenario impact that is also clearly detectable in the late 243 21<sup>st</sup> century means (Fig. 1b). A gradual increase in the liquid FW inflow through Bering 244 Strait over the last decades is consistent with the observed increase (Woodgate, 2018). 245

Net precipitation also shows a clear scenario impact on the shift (Fig. 3). Net precipitation shows complete shift for the CESM LE between the mid 21<sup>st</sup> century and the 247 2090s (and for 90% of the full 40 member CESM LE over a longer period, see Fig. S3c), 248 but only for 18% of the low warming scenario. So while net precipitation over the Arc-250 tic is clearly slowly increasing under both scenarios (Fig. 2), in agreement with obser-251 vations (Peterson et al., 2006; Haine et al., 2015), the changes are small compared to the 252 background variability, at least in the CESM1.1 simulations.

#### 253 4 Discussion

We assessed when a clearly forced change in the Arctic FW budget terms can be 254 detected and found very short emergence periods across all ensemble members in both 255 scenarios in the CESM for the liquid and solid Arctic FW storage. These short emer-256 gence periods suggest a strongly forced change in the Arctic FW storage terms. Hence, 257 based on these CESM results, the large changes in the Arctic FW storage that have been 258 observed over the last three decades are likely to be the beginning of an anthropogenic 259 forced change towards larger liquid FW storage and smaller solid FW storage. In agree-260 ment with another recent study (Wang et al., 2019), we find that the increase in the liq-261 uid FW storage is not exclusively driven by the concurrent decrease in the solid FW stor-262 age: The contribution from the decrease in solid FW storage to the increase in the liq-263

-11-

uid FW storage varies between 35%-89% for the different ensemble members over the
period over which we see shift and emergence in the two storage terms (1996-2015). The
remaining freshening is due to a change in the sum of the FW fluxes, as previously suggested (e.g., Rabe et al., 2014; Carmack et al., 2016). Physically, the storage terms showing emergence prior to the exports means that the changes in the FW exports are an adjustment to the expanded reservoir state.

The finding that the solid FW storage is already showing a forced change agrees 270 with the previous interpretation of the solid FW storage decrease as at least partially 271 driven by climate change due to the loss of sea ice (e.g., Haine et al., 2015; Wang et al., 272 2019; Schweiger et al., 2019). The interpretation of the increased liquid FW storage since 273 the 1990s as a climate change signal is also generally consistent with other climate model 274 studies that show a robust increase of liquid FW storage in the Arctic over the  $21^{st}$  cen-275 tury (e.g., Holland et al., 2006, 2007; Koenigk et al., 2007; Vavrus et al., 2012; Haine et 276 al., 2015; Shu et al., 2018). In particular, the CMIP5 multi-model mean shows an increase 277 in the liquid Arctic FW content since the 1990s, at about 50% of the observed magni-278 tude, suggesting a forced contribution to that change (Shu et al., 2018). However, our 279 results seem to disagree with the interpretation that liquid FW storage changes "observed 280 to date appear to have resulted from natural atmospheric variability" in Johnson et al. 281 (2018). But the two studies may not be in as much in conflict as it appears at first glance. 282 The fact that the reconstructed FW storage timeseries from of Johnson et al. (2018) matches 283 the observed change between the early 1990s and 2012 very well does not preclude the 284 existence of a forced signal in that timeseries, as the FW storage reconstruction is based 285 on the sea level pressure variations from reanalysis, which may contain a climate change 286 signal. This possibility is also alluded to in the recent study of Cornish et al. (2020). Fur-287 thermore, Cornish et al. (2020) find that the strength of the relationship between sea level 288 pressure and liquid FW storage variability varies greatly between different CMIP5 mod-289 els and is weaker than in the model used in Johnson et al. (2018), leaving room for other 290 contributions to the liquid FW content change beside those driven by changes in sea level 291 pressure. Our analysis also does not in any way preclude a contribution from internal 292 variability on top of a forced change. In fact, a contribution from internal variability is 293 likely, and has been shown to exist for the solid FW storage decrease (Notz & Marotzke, 294 2012; Wang et al., 2019; Schweiger et al., 2019). The strong link between the liquid FW 295 storage changes and the sea-level pressure variability identified in Johnson et al. (2018) 296

-12-

may well be part of the physical mechanism that imprints the climate change forcing onto
 the liquid Arctic FW storage.

Similarly to the Arctic FW storage, the short emergence period for the liquid Nares 299 Strait FW flux across all ensemble members and both scenarios suggest that any observed 300 shifts in the 2000s to 2020s towards larger liquid FW fluxes through Nares Strait may 301 already include a climate change driven signal. Hence, it is possible that climate change 302 may have contributed at least partially to the observed larger FW fluxes through Nares 303 Strait between 2003–2006 and 2007–2009 (Münchow, 2016). In fact, Münchow (2016) 304 attributed 69% of this increase in the Nares Strait liquid FW export to the effects of a 305 climate-change driven sea ice decline on the ocean, through the freshening of the sur-306 face waters from ice melt and more efficient momentum transfer from the atmosphere 307 to the ocean under a more mobile ice cover. 308

Since the results presented here are only from one model, which has biases in its 309 representation of the Arctic FW budget, a general agreement with ensembles of CMIP3 310 and CMIP5 models is promising. Different models will, however, likely show differences 311 in the specific years of emergence and shift than those shown here. However, the focus 312 here is not the precise predictions of the shift and emergence years, which fully-coupled 313 models can not provide due to the large impact of internal variability (Deser et al., 2012). 314 Instead, the main take-away should be the overall likelihood of the emergence of forced 315 signals for different budget terms in the presence of large internal variability, and the gen-316 eral timing of the possible emergence of different fluxes within the 21<sup>st</sup> century. 317

Detecting emergence of a climate change signal from the observational timeseries 318 will be more challenging than from model simulations, since even the longer observed 319 FW timeseries only go back to the 1990s (e.g., de Steur et al., 2009; Rabe et al., 2009, 320 2014). Furthermore, rather than a system in steady-state, as in the control simulation, 321 the observations capture a system that is already responding to climate change, in par-322 ticular from the 2000s on (e.g., Kwok, 2018). Comparing the influence of these two com-323 plications present in observed timeseries, we find that it is the changing nature of the 324 FW budget terms during the base period, rather than the much shorter base period it-325 self, that complicates the diagnosis of emergence and shift from observations that, at best, 326 extend to the mid or late 1990s (see section S3 and Fig. S4 for details). 327

-13-

Despite this difficulty with detecting emergence and shift from observations, the 328 current lack of trends in observed liquid FW exports through the Arctic gateways (de 329 Steur et al., 2009; Curry et al., 2014; de Steur et al., 2018; Haine et al., 2015) are con-330 sistent with the CESM results: Full emergence into a new climate state, which would 331 allow a clear trend detection in the presence of increased variability, only occurs for Nares 332 Strait by 2020, and only in some ensemble members but not in others. However, in Nares 333 Strait all ensemble members show emergence before 2030 in the CESM, and by 2042 in 334 Davis Strait. Hence, based on the CESM, we should see sustained increased FW exports 335 in Nares and Davis Strait relatively soon, leading to positive trends eventually. Based 336 on the CESM, we could also soon start to see sustained increased exports and positive 337 trends in Fram Strait, but it could also still take a few decades, depending on the de-338 tails of the natural modes of variability we will experience. This means that continued 339 monitoring of the oceanic fluxes through these gateways and the downstream convec-340 tive regions over the next decades is crucial to record this expected regime shift and as-341 sess its impact on the ocean circulation in the North Atlantic. 342

#### **5** Conclusions

We showed that different Arctic FW budget terms shift toward and emerge into 344 a new climate regime outside their pre-industrial variability at different times. Climate 345 change forced shift and emergence occur first for the Arctic FW storage (both liquid and 346 solid). The simulated emergence period of a climate change signal in both liquid and solid 347 FW storage in the CESM overlaps with the observed increase in Arctic liquid FW stor-348 age between the early 1990s and 2014 and decrease of the solid FW storage since the 1990s 349 (with no Arctic-wide liquid FW data available since 2014). This suggests that the ob-350 served increase in the liquid and solid FW storage is at least partially driven by climate 351 change, rather than the result of unforced internal variability. 352

Generally, oceanic FW fluxes show much earlier emergence than FW fluxes from the atmosphere and land. The first FW flux to emerge is the Nares Strait liquid FW export, which emerges by the end of the 2020s for all members and both scenarios. Emergence in Davis Strait liquid FW export follows Nares Strait, with emergence between the late 2010s and early 2040s. Detecting shift and emergence from shorter timeseries that overlap with the observational period is possible, but the changing nature of many of the FW budget terms, in particular since 2000, can lead to a delayed detection of shift and

-14-

emergence. This means that even if so far there is no trend detected at Nares or Davis 360 Strait in observations, a clearly detectable shift towards positive trends in FW exports 361 is potentially imminent. For Fram Strait the CESM suggests that we are currently in 362 the shift period towards higher liquid FW exports, with the potential for full emergence 363 starting in the mid 2020s. Reduced future emissions may be able to prevent the emer-364 gence into a completely different regime in the late 21<sup>st</sup> century for some FW fluxes such 365 as runoff and the solid Fram Strait FW export, reducing but not avoiding changes in those 366 Arctic FW fluxes. The possibility of ongoing and imminent changes in the oceanic liq-367 uid FW exports highlights the importance of continued observational programs at the 368 Arctic gateways and in the Arctic Ocean, in order to detect these changes in the real world 369 as they occur. 370

#### 371 Acknowledgments

This work was funded by the National Science Foundation under award NSF-OPP 1504348. 372 We thank the NSF-funded FAMOS-2 project and the FAMOS community for provid-373 ing a forum to discuss these results over the years. We also acknowledge the construc-374 tive comments from two anonymous reviewers and the editor on earlier versions of the 375 manuscript. We thank the CESM Large Ensemble and Low-Warming Ensemble projects 376 for making the data freely available to other scientists. Computing resources for the CESM 377 ensembles and the data analysis presented here were provided by the Climate Simula-378 tion Laboratory at NCAR's Computational and Information Systems Laboratory (CISL), 379 sponsored by the National Science Foundation and other agencies. All CESM model out-380 put used is freely available on the NCAR Earth System Grid gateway at https://www 381 .earthsystemgrid.org/dataset/ucar.cgd.ccsm4.lowwarming.html and https:// 382 www.earthsystemgrid.org/dataset/ucar.cgd.ccsm4.CESM\_CAM5\_BGC\_LE.html. The 383 annual mean Arctic FW budget timeseries calculated from this output and analyzed in 384 this paper are archived at the NSF Arctic Data center at https://arcticdata.io/catalog/ 385 view/doi:10.18739/A2CCOTT8X. This manuscript builds upon analysis performed by 386 R. Laiho under the supervision of A. Jahn between 2016 and 2018. A. Jahn conceived 387 the idea for the research, re-analyzed the data, expanded the scope of the analysis, made 388 all final figures, and wrote the manuscript. R. Laiho participated in the editing process 389 for the manuscript. 390

-15-

### 391 References

- Aagaard, K., & Carmack, E. C. (1989). The role of sea ice and other fresh water in
   the Arctic circulation. J. Geophys. Res., 94 (C10), 14485–14498. doi: 10.1029/
   JC094iC10p14485
- Auclair, G., & Tremblay, L. B. (2018). The role of ocean heat transport in rapid sea
   ice declines in the Community Earth System Model Large Ensemble. J. Geo *phys. Res: Oceans*, 123(12), 8941-8957. doi: 10.1029/2018JC014525
- Barnhart, K. R., Miller, C. R., Overeem, I., & Kay, J. E. (2015). Mapping the future expansion of Arctic open water. *Nature Clim. Change*, 6, 280–285. doi:
  10.1038/nclimate2848
- Belkin, I. M., Levitus, S., Antonov, J., & Malmberg, S.-A. (1998). "Great Salinity
   Anomalies" in the North Atlantic. *Prog. Oceanogr.*, 41, 1-68. doi: 10.1016/
   S0079-6611(98)00015-9
- 404 Carmack, E. C., Yamamoto-Kawai, M., Haine, T. W. N., Bacon, S., Bluhm, B. A.,
- Lique, C., ... Williams, W. J. (2016). Freshwater and its role in the Arctic Marine System: Sources, disposition, storage, export, and physical and biogeochemical consequences in the Arctic and global oceans. J. Geophys. Res.:
- <sup>408</sup> Biogeosciences, 121(3), 675-717. doi: 10.1002/2015JG003140
- Cornish, S. B., Kostov, Y., Johnson, H. L., & Lique, C. (2020). Response of Arc tic Freshwater to the Arctic Oscillation in Coupled Climate Models. *Journal of Climate*, 33(7), 2533-2555. doi: 10.1175/JCLI-D-19-0685.1
- Curry, B., Lee, C. M., Petrie, B., Moritz, R. E., & Kwok, R. (2014). Multiyear volume, liquid freshwater, and sea ice transports through Davis Strait, 2004–10.
- 414 J. Phys. Oceanogr., 44(4), 1244-1266. doi: 10.1175/JPO-D-13-0177.1
- de Steur, L., Hansen, E., Gerdes, R., Karcher, M., Fahrbach, E., & Holfort, J.
- (2009). Freshwater fluxes in the East Greenland Current: A decade of observations. *Geophys. Res. Lett.*, 36. doi: 10.1029/2009GL041278
- DeRepentigny, P., Tremblay, L. B., Newton, R., & Pfirman, S. (2016). Patterns of
   sea ice retreat in the transition to a seasonally ice-free Arctic. J. Climate, 29,
   6993-7008. doi: 10.1175/JCLI-D-15-0733.1
- <sup>421</sup> Deser, C., Phillips, A., Bourdette, V., & Teng, H. (2012). Uncertainty in climate
  <sup>422</sup> change projections: the role of internal variability. *Clim. Dyn.*, 38(3), 527-546.
  <sup>423</sup> doi: 10.1007/s00382-010-0977-x

424	de Steur, L., Peralta-Ferriz, C., & Pavlova, O. (2018). Freshwater export in the East
425	Greenland Current freshens the North Atlantic. $Geophys. Res. Lett., 45(24),$
426	13,359-13,366. doi: $10.1029/2018$ GL080207
427	Dickson, R. R., Meincke, J., Malmberg, SA., & Lee, A. J. (1988). The "Great
428	Salinity Anomaly" in the northern North Atlantic 1968–1982. Prog. Oceanogr.,
429	$\mathcal{20}(2),103\text{-}151.$ doi: 10.1016/0079-6611(88)90049-3
430	England, M., Jahn, A., & Polvani, L. (2019). Nonuniform contribution of internal
431	variability to recent Arctic sea ice loss. J. Climate, $32(13)$ , 4039-4053. doi: 10
432	.1175/JCLI-D-18-0864.1
433	Giles, K., Laxon, S., A.L, Ridout, Wingham, D., & Bacon, S. (2012). Western Arc-
434	tic Ocean freshwater storage increased by wind-driven spin-up of the Beaufort
435	Gyre. Nat. Geosci., 5, 194197. doi: 10.1038/NGEO1379
436	Haine, T. W., Curry, B., Gerdes, R., Hansen, E., Karcher, M., Lee, C.,
437	Woodgate, R. (2015). Arctic freshwater export: Status, mechanisms,
438	and prospects. Global and Planetary Change, 125, 13 - 35. doi: 10.1016/
439	j.gloplacha.2014.11.013
440	Hawkins, E., & Sutton, R. (2012). Time of emergence of climate signals. $Geophys.$
441	Res. Lett., $39(1)$ . doi: 10.1029/2011GL050087
442	Holland, M. M., Finnis, J., Barrett, A. P., & Serreze, M. C. (2007). Projected
443	changes in Arctic Ocean freshwater budgets. J. Geophys. Res., 112. doi:
444	10.1029/2006JG000354
445	Holland, M. M., Finnis, J., & Serreze, M. C. (2006). Simulated Arctic Ocean fresh-
446	water budgets in the twentieth and twenty-first centuries. $J.$ Climate, $19(23)$ ,
447	6221–6242. doi: 10.1175/JCLI3967.1
448	Hurrell, J., Holland, M. M., Ghan, P. R. G. S., Kushner, J K. P., Lamarque, JF.,
449	Large, W. G., Marshall, S. (2013). The Community Earth System Model:
450	A Framework for Collaborative Research. Bull. Amer. Meteor. Soc., 94(9),
451	1339-1360. doi: http://dx.doi.org/10.1175/BAMS-D-12-00121.1
452	Jahn, A. (2018). Reduced probability of ice-free summers for 1.5C compared to $2C$
453	warming. Nature Climate Change, $\mathcal{S}(5)$ , 409-413. doi: 10.1038/s41558-018-0127
454	-8
455	Jahn, A., Aksenov, Y., de Cuevas, B., de Steur, L., Häkkinen, S., Hansen, E.,
456	Zhang, J. (2012). Arctic Ocean freshwater - How robust are model simula-

457	tions? J. Geophys. Res., 117. doi: 10.1029/2012JC007907
458	Jahn, A., Kay, J., Holland, M., & Hall, D. (2016). How predictable is the timing of
459	a summer ice-free Arctic? Geophys. Res. Lett., 43, 9113–9120. doi: 10.1002/
460	2016GL070067
461	Johnson, H., Cornish, S., Kostov, Y., Beer, E., & Lique, C. (2018). Arctic Ocean
462	freshwater content and its decadal memory of sea-level pressure. Geophys. Res.
463	Lett., $45(10)$ , 4991-5001. doi: 10.1029/2017GL076870
464	Kay, J. E., Deser, C., Phillips, A., Mai, A., Hannay, C., Strand, G., Vertenstein,
465	M. (2015). The Community Earth System Model (CESM) Large Ensemble
466	Project: A community resource for studying climate change in the presence
467	of internal climate variability. Bull. Amer. Meteor. Soc., 96, 13331349. doi:
468	10.1175/BAMS-D-13-00255.1
469	Koenigk, T., Mikolajewicz, U., Haak, H., & Jungclaus, J. (2007). Arctic freshwa-
470	ter export in the 20th and 21st centuries. J. Geophys. Res., $112$ . doi: 10.1029/
471	2006JG000274
472	Kwok, R. (2018). Arctic sea ice thickness, volume, and multiyear ice coverage: losses
473	and coupled variability (19582018). Environ. Res. Lett., $13$ . doi: 10.1088/1748
474	-9326/aae3ec
475	Kwok, R., Maslowski, W., & Laxon, S. W. (2005). On large outflows of Arctic sea
476	ice into Barents Sea. Geophys. Res. Lett., 32. doi: 10.1029/2005GL024485
477	Lehner, F., Deser, C., & Terray, L. (2017). Toward a new estimate of time of emer-
478	gence of anthropogenic warming: Insights from dynamical adjustment and a
479	large initial-condition model ensemble. J. Climate, $30(19)$ , 7739-7756. doi:
480	10.1175/JCLI-D-16-0792.1
481	Lique, C., Holland, M. M., Dibike, Y. B., Lawrence, D. M., & Screen, J. A. (2016).
482	Modeling the Arctic freshwater system and its integration in the global sys-
483	tem: Lessons learned and future challenges. J. Geophys. Res. Biogeosciences,
484	121(3), 540-566. doi: 10.1002/2015JG003120
485	McPhee, M. G., Proshutinsky, A., Morison, J. H., Steele, M., & Alkire, M. B.
486	(2009). Rapid change in freshwater content of the Arctic Ocean. <i>Geophys.</i>
487	<i>Res. Lett.</i> , 36. doi: 10.1029/2009GL037525
488	Mora, C., Frazier, A. G., Longman, R. J., Dacks, R. S., Walton, M. M., Tong, E. J.,
489	Giambelluca, T. W. (2013). The projected timing of climate departure

490	from recent variability. Nature, $502$ , 183-187. doi: 10.1038/nature12540
491	Morrison, A. L., Kay, J. E., Frey, W. R., Chepfer, H., & Guzman, R. (2019). Cloud
492	response to Arctic sea ice loss and implications for future feedback in the
493	CESM1 Climate Model. J. Geophys. Res: Atmospheres, 124(2), 1003-1020.
494	doi: 10.1029/2018JD029142
495	Münchow, A. (2016). Volume and freshwater flux observations from Nares Strait
496	to the West of Greenland at daily time scales from 2003 to 2009. J. Phys.
497	Oceanogr., 46(1), 141-157. doi: 10.1175/JPO-D-15-0093.1
498	Notz, D., & Marotzke, J. (2012). Observations reveal external driver for Arctic sea-
499	ice retreat. Geophys. Res. Lett., 39. doi: 10.1029/2012GL051094
500	Peterson, B. J., McClelland, J., Curry, R., Holmes, R. M., Walsh, J. E., & Aagaard,
501	K. (2006). Trajectory shifts in the Arctic and subarctic freshwater cycle.
502	Science, $313(5790)$ , 1061–1066. doi: 10.1126/science.1122593
503	Polyakov, I. V., Alexeev, V. A., Belchansky, G. I., Dmitrenko, I. A., Ivanov, V. V.,
504	Kirillov, S. A., Yashayaev, I. (2008). Arctic Ocean freshwater changes
505	over the past 100 years and their causes. J. Climate, 21, 364–384. doi:
506	10.1175/2007JCLI1748.1
507	Prinsenberg, S., & Hamilton, J. (2005). Monitoring the volume, freshwater and heat
508	fluxes passing through Lancaster Sound in the Canadian Arctic Archipelago.
509	AtmosOcean, 43(1), 1-22.
510	Proshutinsky, A., Bourke, R. H., & McLaughlin, F. A. (2002). The role of the Beau-
511	fort Gyre in Arctic climate variability: Seasonal to decadal climate scales. $Geo$ -
512	phys. Res. Lett., 29(23). doi: 10.1029/2002GL015847
513	Proshutinsky, A., & Johnson, M. A. (1997). Two circulation regimes of the wind-
514	driven Arctic Ocean. J. Geophys. Res., $102(C6)$ , 12,493-12,514. doi: 10.1029/
515	97JC00738
516	Proshutinsky, A., Krishfield, R., Timmermans, ML., Toole, J., Carmack, E.,
517	McLaughlin, F., Shimada, K. (2009). Beaufort Gyre freshwater reser-
518	voir: State and variability from observations. J. Geophys. Res.: Oceans, 114.
519	doi: 10.1029/2008JC005104
520	Rabe, B., Karcher, M., Kauker, F., Schauer, U., Toole, J. M., Krishfield, R. A.,
521	Su, J. (2014). Arctic Ocean basin liquid freshwater storage trend 1992–2012.
522	Geophys. Res. Lett., 41(3), 961-968. doi: 10.1002/2013GL058121

523	Rabe, B., Karcher, M., Schauer, U., Toole, J. M., Krishfield, R. A., Pisarevc, S.,
524	Kikuchi, T. (2011). An assessment of Arctic Ocean freshwater content changes
525	from the 1990s to the 2006–2008 period. Deep Sea Res., $58(5)$ , 173–185. doi:
526	10.1016/j.dsr.2010.12.002
527	Rabe, B., Schauer, U., Mackensen, A., Karcher, M., Hansen, E., & Beszczynska-
528	Möller, A. (2009). Freshwater components and transports in the Fram Strait
529	- recent observations and changes since the late 1990s. $Ocean Science, 5(3)$ ,
530	219–233.
531	Sanderson, B., Xu, Y., Tebaldi, C., Wehner, M., O'Neill, B., Jahn, A., Lamar-
532	que, J. (2017). Community Climate Simulations to assess avoided im-
533	pacts in 1.5C and 2C futures. Earth Syst. Dynam., 8, 827–847. doi:
534	10.5194/esd-8-827-2017
535	Schweiger, A. J., Wood, K. R., & Zhang, J. (2019). Arctic sea ice volume variability
536	over 1901–2010: A model-based reconstruction. J. Climate, $32(15)$ , 4731-4752.
537	doi: 10.1175/JCLI-D-19-0008.1
538	Serreze, M. C., Barrett, A. P., Slater, A. G., Woodgate, R. A., Aagaard, K., Lam-
539	mers, R. B., Lee, C. M. (2006). The large-scale freshwater cycle of the
540	Arctic. J. Geophys. Res., 111. doi: 10.1029/2005JC003424
541	Shu, Q., Qiao, F., Song, Z., Zhao, J., & Li, X. (2018). Projected freshening of the
542	Arctic Ocean in the 21st century. J. Geophys. Res: Oceans, 123(12), 9232-
543	9244. doi: 10.1029/2018JC014036
544	Smith, A., & Jahn, A. (2019). Definition differences and internal variability affect
545	the simulated Arctic sea ice melt season. The Cryosphere, $13(1)$ , 1–20. doi: 10
546	.5194/tc-13-1-2019
547	Swart, N. C., Fyfe, J. C., Hawkins, E., Kay, J. E., & Jahn, A. (2015). Influence
548	of internal variability on Arctic sea-ice trends. Nature Clim. Change, 5, 8689.
549	doi: 10.1038/nclimate2483
550	Vavrus, S., Bailey, D., Blazey, B., Holland, M. M., Jahn, A., & Maslanik, J. (2012).
551	The simulation of 21st century Arctic climate in the CCSM4. J. Climate,
552	25(8), 2696–2710. doi: 10.1175/JCLI-D-11-00220.1
553	Wang, Q., Ilicak, M., Gerdes, R., Drange, H., Aksenov, Y., Bailey, D. A., Yeager,
554	S. G. (2016a). An assessment of the Arctic Ocean in a suite of interannual
555	CORE-II simulations. Part II: Liquid freshwater. Ocean Modelling, 99, 86 -

556	109. doi: 10.1016/j.ocemod.2015.12.009
557	Wang, Q., Ilicak, M., Gerdes, R., Drange, H., Aksenov, Y., Bailey, D. A., Yeager,
558	S. G. (2016b). An assessment of the Arctic Ocean in a suite of interannual
559	CORE-II simulations. Part I: Sea ice and solid freshwater. Ocean Modelling,
560	99, 110-132. doi: 10.1016/j.ocemod.2015.12.008
561	Wang, Q., Wekerle, C., Danilov, S., Koldunov, N., Sidorenko, D., Sein, D., Jung,
562	T. (2018). Arctic sea ice decline significantly contributed to the unprecedented
563	liquid freshwater accumulation in the Beaufort Gyre of the Arctic Ocean.
564	Geophys. Res. Lett., $45(10)$ , 4956-4964. doi: 10.1029/2018GL077901
565	Wang, Q., Wekerle, C., Danilov, S., Sidorenko, D., Koldunov, N., Sein, D., Jung,
566	T. (2019). Recent sea ice decline did not significantly increase the total liq-
567	uid freshwater content of the Arctic Ocean. J. Climate, $32(1)$ , 15-32. doi:
568	10.1175/JCLI-D-18-0237.1
569	Woodgate, R. A. (2018). Increases in the Pacific inflow to the Arctic from 1990
570	to 2015, and insights into seasonal trends and driving mechanisms from year-
571	round Bering Strait mooring data. Progress in Oceanography, 160, 124 - 154.
572	doi: 10.1016/j.pocean.2017.12.007
573	Woodgate, R. A., & Aagaard, K. (2005). Revising the Bering Strait freshwater flux
574	into the Arctic Ocean. Geophys. Res. Lett., 32. doi: 10.1029/2004GL021747

#### 575 Supplementary Material

#### <sup>576</sup> Section S1. Climatological Arctic FW Budget in the CESM1.1

The Arctic FW budget is calculated from the CESM1.1 model simulations, rela-577 tive to a reference salinity of 34.8. The surface fluxes (net precipitation and runoff) and 578 the FW storage terms are calculated over the shaded region shown in Fig. 1a, which is 579 delineated by Bering Strait, Barrow Strait, Nares Strait, Fram Strait, and the BSO (shown 580 in Fig. 1a). The liquid FW fluxes through those gateways are given as net FW fluxes 581 over the full depth of the gateways, and for solid FW combine the FW contained in sea 582 ice and in snow on sea ice. The liquid FW storage shown throughout the paper is cal-583 culated down to the 34.8 isohaline, following previous conventions (Serreze et al., 2006; 584 Haine et al., 2015). Davis Strait is shown for reference as an additional strait that is of-585 ten used in Arctic freshwater studies (e.g., Haine et al., 2015; Wang et al., 2016b, 2016a; 586 Shu et al., 2018), but is not part of the Arctic domain over which the FW budget is cal-587 culated. 588

Compared to the observational Arctic FW fluxes for the late 20<sup>th</sup> century, we find 589 that the largest biases in the CESM1.1 FW budget compared to observations are found 590 in the liquid FW exports from the Arctic Ocean. In particular the Fram Strait liquid 591 FW export is much smaller than observed, while the net BSO FW flux is too large. The 592 total solid FW exports on the other hand are slightly too large compared to the obser-593 vations, except in Barrow Strait, where they are too small. However, the net simulated 594 FW export from the Arctic (7066  $\rm km^3/yr$ ) is within the observational uncertainty of the 595 observed net FW export from the Arctic  $(8324\pm1263km^3/yr)$ , so the biases in the fluxes 596 represent a combination of a bias in FW export routes (e.g., more FW export through 597 the BSO, at the expense of the Fram Strait) and a bias between solid and liquid FW ex-598 port (i.e., more solid FW export than observed, at the expense of the liquid FW export), 599 rather than an overall too small FW exchange between the Arctic and North Atlantic. 600 The bias in the liquid versus solid FW fluxes goes along with a larger than observed solid 601 FW storage in the CESM1.1 (see Table S1), indicating that in the late 20<sup>th</sup> century more 602 FW is stored in the solid versus liquid component in the CESM1.1 compared to obser-603 vations. Note that while this means there is more solid FW stored in the CESM1.1 over 604 1980–2000 than observed, the simulated decrease in the solid FW storage over the first 605 decade of the 21<sup>st</sup> century is not too large and agrees well with estimates based on PI-606 OMAS: Haine et al. (2015) found a decrease of  $6,900 \text{ km}^3$  in the solid FW storage based 607

-22-

on PIMOAS between 1980–2000 and 2011, compared to  $6,387 \text{ km}^3$  in the ensemble mean 608 from the CESM1.1 if calculated over the same period and domain as used in Haine et 609 al. (2015) (using Davis Strait rather than Nares and Barrow Straits as boundary west 610 of Greenland; for the smaller Arctic domain used here the simulated decrease over this 611 period is slightly less, at 5,869 km<sup>3</sup>). Hence, the liquid FW storage increase in the CESM1.1 612 over the early 21<sup>st</sup> century that leads to the simulated emergence is not unduly driven 613 by a concurrent too large decline in the solid FW storage over the early 21<sup>st</sup> century pe-614 riod. Eventually, however, the bias in the solid FW storage over the historical period will 615 lead to a too large contribution from sea ice melt, compared to the real world. 616

While there are clear biases in the CESM1.1, it is important to note that limited 617 observations make it challenging to even know what some of the details of the Arctic FW 618 budget should be (as also discussed by Haine et al., 2015; Lique et al., 2016). In partic-619 ular, the liquid FW export west of Greenland has in the past been assumed to be strongly 620 dominated by Barrow Strait/Lancaster Sound, based on the available data at the time 621 (Jahn et al., 2012). However, more recent data from Nares Strait has raised the expected 622 contribution from Nares Strait liquid and solid FW export, due to the inclusion of the 623 surface layer, as well as revealed large, previously unknown interannual variability (Münchow, 624 2016). This new data suggests that the two main channels west of Greenland may in fact 625 be exporting approximately equal amounts of FW from the Arctic (Table S1). Similarly, 626 there is a wide range of estimated solid FW storage (Haine et al., 2015), due to uncer-627 tain Arctic wide sea ice thickness data, in particular prior to the 2000s. 628

# <sup>629</sup> Section S2. IVT Sensitivity to Different Threshold Choices

We here chose an IVT of  $\pm 3.5$  standard deviations, as for normally distributed pro-630 cesses the range between the upper and lower IVT captures 99.95% of values due to un-631 forced internal variability. For most of the FW budget terms, this means that all val-632 ues in the 1800 year long control simulation fall within this  $\pm 3.5$  standard deviation range. 633 However, for a few terms (Fram Strait liquid FW flux, runoff, liquid FW storage, and 634 solid BSO), the IVT threshold is crossed a few times during the 1800 years of the con-635 trol simulation (Fig. S1). Such isolated occurrences outside the  $\pm 3.5$  standard deviation 636 over 1800 years are consistent with the fact that individual very rare (< 0.05% proba-637 bility) events can potentially lead to departures outside the IVT range, even for an IVT 638 range of  $\pm 3.5$  standard deviations. For the BSO solid term, it is also a reflection that 639

this flux is not normally distributed (as it is close to but does not cross the zero line),
so different probabilities apply; however, this term is small and it is only included for
completeness as part of the Arctic FW budget. All results presented also generally hold
if we do not assume normally distributed processes but instead use the maximum and
minimum values of each FW budget terms during the 1800 year long control simulation
plus an extra margin of 10% of the flux to exclude any unsampled rare natural variabilitydriven events (Fig. S3).

As we are using at least 11 ensemble members for the 20<sup>th</sup> and 21<sup>st</sup> centuries, it 647 would be extremely unlikely to see rare events with a probability of <0.05% occurring 648 for all ensemble members over the 181 years of the 20<sup>th</sup> and 21<sup>st</sup> centuries simulation. 649 This means that the detection of spurious complete shifts is highly unlikely. Spurious 650 emergence is not statistically possible, as emergence requires sustained changes outside 651 the pre-industrial IVT range. Hence, this methodology and IVT choice is able to detect 652 truly forced changes in the Arctic FW budget terms. Smaller/larger thresholds than 3.5 653 standard deviations and a non-gaussian approach lead to qualitatively similar results, 654 but some changes in the specific shifts and emergence years due to the change of the prob-655 ability of events outside the chosen range (see Fig. S3). 656

Note that our methodology to determine time of emergence differs from several other 657 "time of emergence" methods (e.g., Hawkins & Sutton, 2012; Mora et al., 2013; Lehner 658 et al., 2017). In particular, we look at annual mean values outside the IVT range rather 659 than considering when the ensemble mean first exceeds the background variability by 660 a certain factor (a typical signal/noise ratio definition of emergence). This approach is 661 most similar to the determination of shift and emergence of Arctic open water days in 662 Barnhart et al. (2015), who demonstrated that there can be substantial differences be-663 tween the emergence time of a variable's ensemble mean versus its unsmoothed trajec-664 tory. As we want to be able to assess when we can expect to observe fluxes and storage 665 that are fully outside the background state, we prefer this time of emergence method-666 ology of using the unsmoothed variables, as that is what we will be able to observe in 667 the real world. 668

# Section S3: Effect of a shorter base period and of sampling a non-steady state system

To provide insights into how shift and emergence detection would look different for 671 observations of the Arctic FW budget, we have repeated our emergence analysis for 20-672 year periods from the control as well as from the historical simulation (Fig. S4). This 673 allows us to assess how the results presented here are affected by using a shorter base 674 period as well as a base period that covers a period where forced changes are starting 675 to affect some of the budget terms. We find that the shorter base period by itself does 676 not affect the main results on emergence, but does change the start and end years by a 677 few years (see Fig. S4b and c versus Fig. S4a). Shift periods on the other hand are more 678 strongly affected by a shorter base period, with some changes of several decades in ei-679 ther direction. This behavior is expected, as emergence detects a sustained, forced change 680 while shift is triggered by an individual event, so a small change in the IVT will affect 681 shift more strongly than emergence. Sampling a non-steady state system for 20 years, 682 however, has a big effect on detecting emergence. Emergence patterns similar to the ones 683 based on the full length of the control simulation are found primarily for a 20 year pe-684 riod from the historical simulation that ends before 2000 (see Fig. S4a, d, g). Once the 685 base period extends past 2000, emergence is reached later, in particular for the terms 686 that show early emergence (Fig. S4e, f, h, i). Nonetheless, the general order of emergence 687 of FW budget terms remains the same even for base periods that extend to 2009. For 688 base periods that extend past 2009, however, even the order of emergence changes, as 689 the base period from 2000-2019 now samples the already very different FW storage terms, 690 leading to a much later emergence of these terms compared to their already very differ-691 ent base state. Hence, it is the changing nature of the FW budget terms during the base 692 period, rather than the much shorter base period itself, that complicates the diagnosis 693 of emergence and shift from observations that, at best, extend to the mid or late 1990s. 694

#### 695 References

Aagaard, K., & Carmack, E. C. (1989). The role of sea ice and other fresh water in
 the Arctic circulation. J. Geophys. Res., 94 (C10), 14485–14498. doi: 10.1029/
 JC094iC10p14485

Auclair, G., & Tremblay, L. B. (2018). The role of ocean heat transport in rapid sea ice declines in the Community Earth System Model Large Ensemble. J. Geo-

-25-

701	phys. Res: Oceans, 123(12), 8941-8957. doi: 10.1029/2018JC014525
702	Barnhart, K. R., Miller, C. R., Overeem, I., & Kay, J. E. (2015). Mapping the fu-
703	ture expansion of Arctic open water. Nature Clim. Change, $6$ , 280–285. doi:
704	10.1038/nclimate 2848
705	Belkin, I. M., Levitus, S., Antonov, J., & Malmberg, SA. (1998). "Great Salinity
706	Anomalies" in the North Atlantic. Prog. Oceanogr., 41, 1-68. doi: 10.1016/
707	S0079-6611(98)00015-9
708	Carmack, E. C., Yamamoto-Kawai, M., Haine, T. W. N., Bacon, S., Bluhm, B. A.,
709	Lique, C., Williams, W. J. (2016). Freshwater and its role in the Arctic
710	Marine System: Sources, disposition, storage, export, and physical and bio-
711	geochemical consequences in the Arctic and global oceans. J. Geophys. Res.:
712	Biogeosciences, 121(3), 675-717. doi: 10.1002/2015JG003140
713	Cornish, S. B., Kostov, Y., Johnson, H. L., & Lique, C. (2020). Response of Arc-
714	tic Freshwater to the Arctic Oscillation in Coupled Climate Models. Journal of
715	Climate, 33(7), 2533-2555. doi: 10.1175/JCLI-D-19-0685.1
716	Curry, B., Lee, C. M., Petrie, B., Moritz, R. E., & Kwok, R. (2014). Multiyear vol-
717	ume, liquid freshwater, and sea ice transports through Davis Strait, 2004–10.
718	J. Phys. Oceanogr., 44(4), 1244-1266. doi: 10.1175/JPO-D-13-0177.1
719	de Steur, L., Hansen, E., Gerdes, R., Karcher, M., Fahrbach, E., & Holfort, J.
720	(2009). Freshwater fluxes in the East Greenland Current: A decade of ob-
721	servations. Geophys. Res. Lett., 36. doi: 10.1029/2009GL041278
722	DeRepentigny, P., Tremblay, L. B., Newton, R., & Pfirman, S. (2016). Patterns of
723	sea ice retreat in the transition to a seasonally ice-free Arctic. $J.$ Climate, 29,
724	6993–7008. doi: $10.1175/JCLI$ -D-15-0733.1
725	Deser, C., Phillips, A., Bourdette, V., & Teng, H. (2012). Uncertainty in climate
726	change projections: the role of internal variability. Clim. Dyn., 38(3), 527-546.
727	doi: 10.1007/s00382-010-0977-x
728	de Steur, L., Peralta-Ferriz, C., & Pavlova, O. (2018). Freshwater export in the East
729	Greenland Current freshens the North Atlantic. $Geophys. Res. Lett., 45(24),$
730	13,359-13,366. doi: $10.1029/2018$ GL080207
731	Dickson, R. R., Meincke, J., Malmberg, SA., & Lee, A. J. (1988). The "Great
732	Salinity Anomaly" in the northern North Atlantic 1968–1982. Prog. Oceanogr.,
733	20(2), 103-151. doi: $10.1016/0079-6611(88)90049-3$

734	England, M., Jahn, A., & Polvani, L. (2019). Nonuniform contribution of internal
735	variability to recent Arctic sea ice loss. J. Climate, $32(13)$ , 4039-4053. doi: 10
736	.1175/JCLI-D-18-0864.1
737	Giles, K., Laxon, S., A.L, Ridout, Wingham, D., & Bacon, S. (2012). Western Arc-
738	tic Ocean freshwater storage increased by wind-driven spin-up of the Beaufort
739	Gyre. Nat. Geosci., 5, 194197. doi: 10.1038/NGEO1379
740	Haine, T. W., Curry, B., Gerdes, R., Hansen, E., Karcher, M., Lee, C.,
741	Woodgate, R. (2015). Arctic freshwater export: Status, mechanisms,
742	and prospects. Global and Planetary Change, 125, 13 - 35. doi: 10.1016/
743	j.gloplacha.2014.11.013
744	Hawkins, E., & Sutton, R. (2012). Time of emergence of climate signals. $Geophys.$
745	Res. Lett., $39(1)$ . doi: 10.1029/2011GL050087
746	Holland, M. M., Finnis, J., Barrett, A. P., & Serreze, M. C. (2007). Projected
747	changes in Arctic Ocean freshwater budgets. J. Geophys. Res., 112. doi:
748	10.1029/2006 JG000354
749	Holland, M. M., Finnis, J., & Serreze, M. C. (2006). Simulated Arctic Ocean fresh-
750	water budgets in the twentieth and twenty-first centuries. $J.$ Climate, $19(23)$ ,
751	6221–6242. doi: 10.1175/JCLI3967.1
752	Hurrell, J., Holland, M. M., Ghan, P. R. G. S., Kushner, J K. P., Lamarque, JF.,
753	Large, W. G., Marshall, S. (2013). The Community Earth System Model:
754	A Framework for Collaborative Research. Bull. Amer. Meteor. Soc., 94(9),
755	1339-1360. doi: http://dx.doi.org/10.1175/BAMS-D-12-00121.1
756	Jahn, A. (2018). Reduced probability of ice-free summers for 1.5C compared to $2C$
757	warming. Nature Climate Change, $\mathcal{S}(5)$ , 409-413. doi: 10.1038/s41558-018-0127
758	-8
759	Jahn, A., Aksenov, Y., de Cuevas, B., de Steur, L., Häkkinen, S., Hansen, E.,
760	Zhang, J. (2012). Arctic Ocean freshwater - How robust are model simula-
761	tions? J. Geophys. Res., 117. doi: 10.1029/2012JC007907
762	Jahn, A., Kay, J., Holland, M., & Hall, D. (2016). How predictable is the timing of
763	a summer ice-free Arctic? Geophys. Res. Lett., 43, 9113–9120. doi: 10.1002/
764	2016GL070067
765	Johnson, H., Cornish, S., Kostov, Y., Beer, E., & Lique, C. (2018). Arctic Ocean
766	freshwater content and its decadal memory of sea-level pressure. Geophys. Res.

767	Lett., $45(10)$ , 4991-5001. doi: 10.1029/2017GL076870
768	Kay, J. E., Deser, C., Phillips, A., Mai, A., Hannay, C., Strand, G., Vertenstein,
769	M. (2015). The Community Earth System Model (CESM) Large Ensemble
770	Project: A community resource for studying climate change in the presence
771	of internal climate variability. Bull. Amer. Meteor. Soc., 96, 13331349. doi:
772	10.1175/BAMS-D-13-00255.1
773	Koenigk, T., Mikolajewicz, U., Haak, H., & Jungclaus, J. (2007). Arctic freshwa-
774	ter export in the 20th and 21st centuries. J. Geophys. Res., $112$ . doi: 10.1029/
775	2006JG000274
776	Kwok, R. (2018). Arctic sea ice thickness, volume, and multiyear ice coverage: losses
777	and coupled variability (19582018). Environ. Res. Lett., 13. doi: 10.1088/1748
778	-9326/aae3ec
779	Kwok, R., Maslowski, W., & Laxon, S. W. (2005). On large outflows of Arctic sea
780	ice into Barents Sea. Geophys. Res. Lett., 32. doi: 10.1029/2005GL024485
781	Lehner, F., Deser, C., & Terray, L. (2017). Toward a new estimate of time of emer-
782	gence of anthropogenic warming: Insights from dynamical adjustment and a
783	large initial-condition model ensemble. J. Climate, $30(19)$ , 7739-7756. doi:
784	10.1175/JCLI-D-16-0792.1
785	Lique, C., Holland, M. M., Dibike, Y. B., Lawrence, D. M., & Screen, J. A. (2016).
786	Modeling the Arctic freshwater system and its integration in the global sys-
787	tem: Lessons learned and future challenges. J. Geophys. Res. Biogeosciences,
788	121(3), 540-566. doi: 10.1002/2015JG003120
789	McPhee, M. G., Proshutinsky, A., Morison, J. H., Steele, M., & Alkire, M. B.
790	(2009). Rapid change in freshwater content of the Arctic Ocean. <i>Geophys.</i>
791	Res. Lett., 36. doi: 10.1029/2009GL037525
792	Mora, C., Frazier, A. G., Longman, R. J., Dacks, R. S., Walton, M. M., Tong, E. J.,
793	Giambelluca, T. W. (2013). The projected timing of climate departure
794	from recent variability. Nature, $502$ , 183-187. doi: 10.1038/nature12540
795	Morrison, A. L., Kay, J. E., Frey, W. R., Chepfer, H., & Guzman, R. (2019). Cloud
796	response to Arctic sea ice loss and implications for future feedback in the
797	CESM1 Climate Model. J. Geophys. Res: Atmospheres, 124(2), 1003-1020.
798	doi: 10.1029/2018JD029142
799	Münchow, A. (2016). Volume and freshwater flux observations from Nares Strait

-28-

800	to the West of Greenland at daily time scales from 2003 to 2009. J. Phys.
801	Oceanogr., 46(1), 141-157. doi: 10.1175/JPO-D-15-0093.1
802	Notz, D., & Marotzke, J. (2012). Observations reveal external driver for Arctic sea-
803	ice retreat. Geophys. Res. Lett., 39. doi: 10.1029/2012GL051094
804	Peterson, B. J., McClelland, J., Curry, R., Holmes, R. M., Walsh, J. E., & Aagaard,
805	K. (2006). Trajectory shifts in the Arctic and subarctic freshwater cycle.
806	Science, $313(5790)$ , 1061–1066. doi: 10.1126/science.1122593
807	Polyakov, I. V., Alexeev, V. A., Belchansky, G. I., Dmitrenko, I. A., Ivanov, V. V.,
808	Kirillov, S. A., Yashayaev, I. (2008). Arctic Ocean freshwater changes
809	over the past 100 years and their causes. J. Climate, 21, 364–384. doi:
810	10.1175/2007JCLI1748.1
811	Prinsenberg, S., & Hamilton, J. (2005). Monitoring the volume, freshwater and heat
812	fluxes passing through Lancaster Sound in the Canadian Arctic Archipelago.
813	AtmosOcean, 43(1), 1-22.
814	Proshutinsky, A., Bourke, R. H., & McLaughlin, F. A. (2002). The role of the Beau-
815	fort Gyre in Arctic climate variability: Seasonal to decadal climate scales. $Geo$ -
816	phys. Res. Lett., $29(23)$ . doi: 10.1029/2002GL015847
817	Proshutinsky, A., & Johnson, M. A. (1997). Two circulation regimes of the wind-
818	driven Arctic Ocean. J. Geophys. Res., $102(C6)$ , 12,493-12,514. doi: 10.1029/
819	97JC00738
820	Proshutinsky, A., Krishfield, R., Timmermans, ML., Toole, J., Carmack, E.,
821	McLaughlin, F., Shimada, K. (2009). Beaufort Gyre freshwater reser-
822	voir: State and variability from observations. J. Geophys. Res.: Oceans, 114.
823	doi: 10.1029/2008JC005104
824	Rabe, B., Karcher, M., Kauker, F., Schauer, U., Toole, J. M., Krishfield, R. A.,
825	Su, J. (2014). Arctic Ocean basin liquid freshwater storage trend 1992–2012.
826	Geophys. Res. Lett., 41(3), 961-968. doi: 10.1002/2013GL058121
827	Rabe, B., Karcher, M., Schauer, U., Toole, J. M., Krishfield, R. A., Pisarevc, S.,
828	Kikuchi, T. (2011). An assessment of Arctic Ocean freshwater content changes
829	from the 1990s to the 2006–2008 period. Deep Sea Res., $58(5)$ , 173–185. doi:
830	10.1016/j.dsr.2010.12.002
831	Rabe, B., Schauer, U., Mackensen, A., Karcher, M., Hansen, E., & Beszczynska-
832	Möller, A. (2009). Freshwater components and transports in the Fram Strait

833	- recent observations and changes since the late 1990s. $Ocean Science, 5(3)$ ,
834	219–233.
835	Sanderson, B., Xu, Y., Tebaldi, C., Wehner, M., O'Neill, B., Jahn, A., Lamar-
836	que, J. (2017). Community Climate Simulations to assess avoided im-
837	pacts in 1.5C and 2C futures. Earth Syst. Dynam., 8, 827–847. doi:
838	10.5194/esd-8-827-2017
839	Schweiger, A. J., Wood, K. R., & Zhang, J. (2019). Arctic sea ice volume variability
840	over 1901–2010: A model-based reconstruction. J. Climate, $32(15)$ , 4731-4752.
841	doi: 10.1175/JCLI-D-19-0008.1
842	Serreze, M. C., Barrett, A. P., Slater, A. G., Woodgate, R. A., Aagaard, K., Lam-
843	mers, R. B., $\dots$ Lee, C. M. (2006). The large-scale freshwater cycle of the
844	Arctic. J. Geophys. Res., 111. doi: 10.1029/2005JC003424
845	Shu, Q., Qiao, F., Song, Z., Zhao, J., & Li, X. (2018). Projected freshening of the
846	Arctic Ocean in the 21st century. J. Geophys. Res: Oceans, 123(12), 9232-
847	9244. doi: 10.1029/2018JC014036
848	Smith, A., & Jahn, A. (2019). Definition differences and internal variability affect
849	the simulated Arctic sea ice melt season. The Cryosphere, $13(1)$ , 1–20. doi: 10
850	.5194/tc-13-1-2019
851	Swart, N. C., Fyfe, J. C., Hawkins, E., Kay, J. E., & Jahn, A. (2015). Influence
852	of internal variability on Arctic sea-ice trends. Nature Clim. Change, 5, 8689.
853	doi: 10.1038/nclimate2483
854	Vavrus, S., Bailey, D., Blazey, B., Holland, M. M., Jahn, A., & Maslanik, J. (2012).
855	The simulation of 21st century Arctic climate in the CCSM4. J. Climate,
856	25(8), 2696-2710. doi: 10.1175/JCLI-D-11-00220.1
857	Wang, Q., Ilicak, M., Gerdes, R., Drange, H., Aksenov, Y., Bailey, D. A., Yeager,
858	S. G. (2016a). An assessment of the Arctic Ocean in a suite of interannual
859	CORE-II simulations. Part II: Liquid freshwater. Ocean Modelling, 99, 86 -
860	109. doi: 10.1016/j.ocemod.2015.12.009
861	Wang, Q., Ilicak, M., Gerdes, R., Drange, H., Aksenov, Y., Bailey, D. A., Yeager,
862	S. G. (2016b). An assessment of the Arctic Ocean in a suite of interannual
863	CORE-II simulations. Part I: Sea ice and solid freshwater. Ocean Modelling,
864	99, 110-132. doi: 10.1016/j.ocemod.2015.12.008
865	Wang, Q., Wekerle, C., Danilov, S., Koldunov, N., Sidorenko, D., Sein, D., Jung,

866	T. (2018). Arctic sea ice decline significantly contributed to the unprecedented
867	liquid freshwater accumulation in the Beaufort Gyre of the Arctic Ocean.
868	Geophys. Res. Lett., $45(10)$ , 4956-4964. doi: 10.1029/2018GL077901
869	Wang, Q., Wekerle, C., Danilov, S., Sidorenko, D., Koldunov, N., Sein, D., Jung,
870	T. (2019). Recent sea ice decline did not significantly increase the total liq-
871	uid freshwater content of the Arctic Ocean. J. Climate, $32(1)$ , 15-32. doi:
872	10.1175/JCLI-D-18-0237.1
873	Woodgate, R. A. (2018). Increases in the Pacific inflow to the Arctic from 1990
874	to 2015, and insights into seasonal trends and driving mechanisms from year-
875	round Bering Strait mooring data. Progress in Oceanography, 160, 124 - 154.
876	doi: 10.1016/j.pocean.2017.12.007
877	Woodgate, R. A., & Aagaard, K. (2005). Revising the Bering Strait freshwater flux

into the Arctic Ocean. Geophys. Res. Lett., 32. doi:  $10.1029/2004 {\rm GL}021747$ 

878

Climatological Arctic Ocean freshwater (FW) budget (1980-2000). Ob-Table S1. servational values are partially taken from the compilation by Serreze et al. (2006) (indicated by \* in the table). Terms that are not from Serreze et al. (2006) are: Bering Strait solid FW fluxes (Woodgate & Aagaard, 2005), BSO solid FW fluxes (Kwok et al., 2005), Nares Strait liquid and solid FW fluxes (Münchow, 2016), Barrow Strait solid and liquid fluxes (Prinsenberg & Hamilton, 2005), and Davis Strait solid and liquid FW fluxes (Haine et al., 2015). The solid FW storage in the Arctic is shown as range, based on the values given in Serreze et al. (2006) and Haine et al. (2015). These two estimates differ in the assumed mean ice thickness (thinner ice assumed in Serreze et al. (2006) than Haine et al. (2015)) as well as in their Arctic domain, with the Arctic domain in Serreze et al. (2006) smaller than our domain (entirely excluding the CAA) and the domain in Haine et al. (2015) larger than our domain (including Baffin Bay down to Davis Strait). In the CESM1.1, the impact of these domain differences compared to the Arctic domain used here is an additional solid FW storage of 1,868 km3 for the domain of Haine et al. (2015) and  $2988 \text{ km}^3$  less solid FW storage for the domain of Serreze et al. (2006), which does not change the fact that the CESM1.1 has too much solid FW storage. However, note that the solid FW flux and storage includes FW from the snow on sea ice as well as from the ice itself while the observational estimates typically only include the FW in the sea ice, which leads to a difference of about 10%. Further note that the Nares and Barrow Strait values are from the early 2000s, rather than the late 20<sup>th</sup> century, as no earlier data exists. If available, error estimates for the observations are included. Model values show the ensemble mean values, and the  $\pm$  indicates the standard deviation of the 40-member CESM LE in the 21-yr averages. All FW fluxes are quoted in km<sup>3</sup>/year, and the FW storage is quoted in km<sup>3</sup>. All values are annual mean net fluxes, for oceanic fluxes over the full depth of each channel, combining negative and positive fluxes through a strait, where applicable. Positive values indicate FW sources and negative values indicate FW sinks for the Arctic Ocean. Note that Davis Strait is included here for reference only, with the surface fluxes and storage calculated over the Arctic Ocean domain delineated by Nares Strait and Barrow Strait west of Greenland (see Fig. 1a).

FW fluxes	Observations	CESM LE
River runoff	$3200 \pm 110^*$	$3358 \pm 55$
Net precipitation	$2000 \pm 200^{*}$	$1958 \pm 32$
Bering Strait liquid FW	$2400 \pm 300^*$	$2159{\pm}66$
Bering Strait solid FW	$140\ \pm40$	$56{\pm}14$
Barrow Strait liquid FW	-1510	$-567 \pm 25$
Barrow Strait solid FW	-76	$2\pm1$
Nares Strait liquid FW	$-1356 \pm 236$	$-1439 \pm 69$
Nares Strait solid FW	$-252 \pm 63$	$-395{\pm}15$
Davis Strait liquid FW	$-3200 \pm 320$	$-2044{\pm}69$
David Strait solid FW	-160	$-701\pm24$
Fram Strait liquid FW	$-2700 \pm 530^{*}$	$-948 \pm 68$
Fram Strait solid FW	$-2300 \pm 340^{*}$	$-2776 \pm 174$
BSO liquid FW	$-90 \pm 94^{*}$	$-852{\pm}50$
BSO solid FW	-40	$-91{\pm}41$
Liquid FW storage	$74,000 \pm 7400^*$	$77,485 \pm 1562$
Solid FW storage	$10,000^{*} - 17,800$	$21,931{\pm}1011$



Figure S1. Variability in the control simulation. The  $\pm 3.5$  standard deviation threshold for each variable (which is used to determine shift and emergence in the 20<sup>th</sup> and 21<sup>st</sup> centuries simulations) is shown as solid dark grey lines. Flux terms (a-n) and storage terms (o, p) are labeled in the panels. Note that all flux panels (a-n) and all storage panels (o-p) each have the same y-axis range, but that the axis are offset from each other.



Figure S2. Small FW budget terms over time. As in Fig. 2, but for the small (less than  $300 \text{ km}^3/\text{yr}$  in the observed net fluxes) FW fluxes not shown in Fig. 2. Note that the y-axis is the same for all panels, but is different from Fig. 2 to allow a more meaningful depiction of these small fluxes.



Figure S3. Sensitivity of results to different IVT choices. As Fig. 3, but for an IVT defined as (a) the maximum/minmum values in the control  $\pm 10\%$  of the mean, (b)  $\pm 3$  standard deviations, (c)  $\pm 3.5$  standard deviations, and (d)  $\pm 4$  standard deviations. This figure also includes the FW fluxes with a net observed flux of less than 300 km<sup>3</sup>/yr, which were not shown in Fig. 3. These different IVT choices (a, b, d) show qualitatively similar results as for 3.5 standard deviations (c), with the largest changes primarily in the start dates of the shift periods, due to the smaller/larger IVT range. None of the main conclusions are affected by the choice of the IVT, as they mainly focus on the emergence of the forced signal.



Figure S4. Influence of different base periods on shift and emergence. Shift and emergence, as shown in Fig. 3, but using different base periods in the different panels to determine the IVT, to assess the impact of a shorter base period, as would be available from observations. (a) Using the full 1800 years of the control simulation, same as Fig. 3, (b) using only 20 years of the control (here years 1000-1019), (c) using a different set of 20 years from the control (here years 400-419), (d/g) using years 1980-1999 from ensemble member 1/10 of the CESM LE, (e/h) using years 1990–2009 from ensemble member 1/10, (f/i) using years 2000-2019 from ensemble member 1/10. This shows that sampling of a system more and more affected by climate change if years after 2000 are included in the base period affects the results more than using a shorter base period, in particular for emergence (with shift sensitive to both). Results are similar for 30 year instead of 20 year periods. Members 1 and 10 are shown in panels d–f and g–i, respectively, to illustrate the effects of sampling different 20-yr periods under the same external forcing but with different internal variability. Other ensemble members show slightly different patterns, but changes are qualitatively similar to the difference between the two members shown here.